Author's response to reviews

Title: Estimation of the benchmark dose of urinary cadmium as the reference level for renal dysfunction: A large sample study in five cadmium polluted areas in China

Authors:

Shen Ke (shke@bjtu.edu.cn)
Xi-Yu Cheng (xycheng@bjtu.edu.cn)
Jie-Ying Zhang (13121589@bjtu.edu.cn)
Wen-Jing Jia (12121628@bjtu.edu.cn)
Hao Li (12121666@bjtu.edu.cn)
Hui-Fang Luo (1451316918@qq.com)
Peng-He Ge (phge1000@126.com)
Ze-Min Liu (zmliu1000@163.com)
Hong-Mei Wang (hmwang1000@163.com)
Jin-Sheng He (jshhe@bjtu.edu.cn)
Zhi-Nan Chen (chen_zhinan56@163.com)

Version: 4 Date: 17 June 2015

Author's response to reviews: see over
Dear editor,

Thank you very much for your kind consideration. The figure in our manuscript (MS: 4539679531541518) is from a website and it is difficult to obtain the official proof of permission. We have already described the distribution of the sampling areas in the text and the Figure 1 is not essential. Therefore, we have removed the figure from the revised MS and the corresponding description in the text has been slightly revised. All changes in the revised manuscript are marked in red.

We thank you again for your kind and helpful suggestions for our manuscript.

Sincerely

Zhi-Nan Chen

Response to reviewers/Editor's Comments:

1) Image attribution
Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.
If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.
Response: This figure is from a website and it is difficult to obtain the official proof of permission. We have already described the distribution of the sampling areas in the text and the Figure 1 is not essential. Therefore, we have removed the figure from the revised MS and the corresponding description in the text has been slightly revised (Page 6, Line 107-109).